
Shift summary
14-21/03

LILIANA CONGEDO AND GIULIANA GALATI



System status

• Stability run 206 interrupted on Tuesday, since from about 
1am the current of all detectors increased (see next slide).
All detectors were switched OFF.

• HFO bottle finished yesterday. It was replaced at about 3 pm
and the RPCs flushed up to this morning.

• Today at about 10 the source has been switched OFF and 
two HV scans have been performed: 
ID 725 and 726

• The source has just been switched ON (ABS 2.2) and a new 
stability run with eco2 has been started (ID 207).
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Current increase on Tuesday
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Small peaks in all fluxes and wet MFC at about 1am
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